Author’s response to reviews

Title: Effects of Body Weight and Alcohol Consumption on Insulin Sensitivity

Authors:

Qiwei X Paulson (xiaqw@mail.utexas.edu)
Jina Hong (j_hong@mail.utexas.edu)
Valerie B Holcomb (valerieholcomb@mail.utexas.edu)
Nomeli P Nunez (nomeli@mail.utexas.edu)

Version: 6 Date: 11 March 2010

Author’s response to reviews: see over
The following suggested modifications were made to the manuscript:

Major revisions (we require the author to make these changes)

1. Competing interests: Please include a `Competing interests¿ section between the Conclusions and Authors¿ contributions. If there are none to declare, please write `The authors declare that they have no competing interests¿.

   This has been done as suggested.

2. Authors¿ contributions: Please place an Authors¿ contributions section towards the end of the manuscript text, directly preceding the Acknowledgements. Use the following format as an example, including using initials to refer to each author¿s contribution:

   This section has been modified.

3. Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

   Tables are not linked files---they were fixed.

4. Figures: The figures are quite large; the maximum size for each individual figure is A4. If your figure needs to remain larger than this and cannot be reduced we recommend you separate it into component parts and upload these separately. Please remember to update the figure titles and references within the text. Figures: Please reduce the volume of text presented in the figure legends to no more than 250 words or it will compete with the image for space in the final publication.

   Figures and figure legends were modified.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

5. Line numbers: Please remove these

   Lines were removed

6. Footer: Please remove the footer. We cannot support headers and footers, and the information in your footer is already listed in the `Acknowledgments¿ section.
7. Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalize only the first word and proper nouns, and do not include full stops/periods. For example, `The Title of My Prestigious Publication: A Scientific Study.¿ should be changed to `The title of my prestigious publication: a scientific study¿

These modifications were done

8. Email address: please include email addresses for all the authors on the title page, using the following format:

E-mails are included now

9. Title page: Please remove the following:
   Figure count
   Tables count
   `Supplementary online material¿ heading
   Running title
   Key words

   They were removed.

10. Table headings: Please remove the repeated column headings from where the tables run over two pages.

   This was fixed

11. Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both)

   We made this modification.

12. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

   We checked the manuscript for Typography